\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at84a.txt}{\rawdatadir/at84a.txt}~Benedick (1986)~\cite{w_bene86}
\W\label{at84a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at84a~\protect\cite[Benedick (1986)]{w_bene86}\label{at84a}}\\
\T\input{\datadir/at84a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H4 \\
Sub-Category: & spherical, high explosive & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & .6-3.0 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at109a.txt}{\rawdatadir/at109a.txt}~Bull (1978)~\cite{d_bull78}
\W\label{at109a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at109a~\protect\cite[Bull (1978)]{d_bull78}\label{at109a}}\\
\T\input{\datadir/at109a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H4 \\
Sub-Category: & spherical, high explosive & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at109b.txt}{\rawdatadir/at109b.txt}~Bull (1979)~\cite{d_bull79a}
\W\label{at109b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at109b~\protect\cite[Bull (1979)]{d_bull79a}\label{at109b}}\\
\T\input{\datadir/at109b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H4 \\
Sub-Category: & spherical, high explosive & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 0.8 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at109c.txt}{\rawdatadir/at109c.txt}~Hikita (1975)~\cite{t_hiki75}
\W\label{at109c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at109c~\protect\cite[Hikita (1975)]{t_hiki75}\label{at109c}}\\
\T\input{\datadir/at109c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H4 \\
Sub-Category: & spherical, high explosive & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1.2 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at79b.txt}{\rawdatadir/at79b.txt}~Matsui (1979)~\cite{h_mats79}
\W\label{at79b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at79b~\protect\cite[Matsui (1979)]{h_mats79}\label{at79b}}\\
\T\input{\datadir/at79b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H4 \\
Sub-Category: & spherical & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1.33 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at32b.txt}{\rawdatadir/at32b.txt}~Murray (1981)~\cite{s_murr81}
\W\label{at32b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at32b~\protect\cite[Murray (1981)]{s_murr81}\label{at32b}}\\
\T\input{\datadir/at32b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H4 \\
Sub-Category: & high explosive & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

